Author's response to reviews

Title: A retrospective study showing Maintenance treatment options for paediatric CD in the first year following diagnosis after induction of remission with EEN: supplemental enteral nutrition is better than nothing!

Authors:

Hazel Duncan (hazeljduncan@gmail.com)
Elaine Buchanan (elaine.buchanan@ggc.scot.nhs.uk)
Tracey Cardigan (tracey.cardigan@ggc.scot.nhs.uk)
Victoria Garrick (vgarrick@nhs.net)
Lee Curtis (leecurtis@nhs.net)
Paraic McGrogan (Paraic.McGrogan@ggc.scot.nhs.uk)
Andrew Barclay (Andrew.Barclay@ggc.scot.nhs.uk)
Richard K Russell (richardrussell@nhs.net)

Version: 4
Date: 2 March 2014

Author's response to reviews:

Response to reviewers in attached letter.